Author's response to reviews

Title: Clinical significance in the number of involved lymph nodes in patients that underwent surgery for pathological stage III-N2 non-small cell lung cancer

Authors:

Takeshi Hanagiri (hanagiri@med.uoeh-u.ac.jp)
Masaru Takenaka (masarururu_1230@yahoo.co.jp)
Soich Oka (okasou@med.uoeh-u.ac.jp)
Yoshiki Shigematsu (yoshiki@med.uoeh-u.ac.jp)
Yoshika Nagata (y-nagata@med.uoeh-u.ac.jp)
Hidehiko Shimokawa (mikegonn@yahoo.co.jp)
Hidetaka Uramoto (hidetaka@med.uoeh-u.ac.jp)
Fumihiro Tanaka (tanaka@med.uoeh-u.ac.jp)

Version: 4 Date: 21 October 2011

Author's response to reviews:

Tables was created using a Tables tool, using MS Word auto-format template: simple 1